Author's response to reviews

Title: A qualitative study of nursing student experiences of clinical practice

Authors:

Farkhondeh Sharif (shariffarkhondeh@hotmail.com)
Sara Masoumi (sarah_mas25@yahoo.com)

Version: 2 Date: 29 August 2005

Author's response to reviews: see over
2005-08-29
Answer to The Reviewer's report
Reviewer: Dr Ella McAfferty
Dear Ella
Thank you very much for your time and advice. I added an explanation about observer and the purpose of using focus group. The demographic characteristics of sample in the table were extra and it has been deleted, but I added a sentence at the beginning of the result section. I tried my best. I wish you success.

Reviewer: Dr Elizabeth Severinsson
Dear Elizabeth
Thank you so much for your time and your advice. Thank you very much for introducing your related articles to me which was very useful. Also I did further literature search. Answer to questions:
1. Heading named data analysis has been added.
2. Actually I did qualitative content analysis an you mentioned. creation of code, categories and themes according to Graneheim and Lundman is the core feature of qualitative content analysis which I did in my research. I did some changes in figure 1 to be more clear. analysis.
3. Initial clinical anxiety is one of the 4 main themes 4. Behyar has been changed to Registered Practical Nurse because it is the same. I did the other suggestions which you mentioned. My revised manuscript is uploaded.

With best wishes
Farkhondeh Sharif.